

TECHNOLOGICAL and ECONOMIC DEVELOPMENT of ECONOMY

2024 Volume 30 Issue 6

Pages 1618-1651

https://doi.org/10.3846/tede.2024.21858

WELL-BEING EFFECTS OF THE DIGITAL PLATFORM ECONOMY: THE CASE OF TEMPORARY AND SELF-EMPLOYMENT

Maite BLÁZQUEZ¹, Ainhoa HERRARTE¹, Ana I. MORO-EGIDO²

¹Departmento de Análisis Económico: Teoría Económica e Historia Económica, Facultad de Ciencias

Económicas y Empresariales, Universidad Autónoma de Madrid, Madrid, Spain

²Departmento de Teoría e Historia Económica, Facultad de Ciencias Económicas y Empresariales,

Universidad de Granada, Granada, Spain

 received 7 May 2023 accepted 6 March 2024 first published online 17 September 2024 	Abstract. The surge in non-traditional employment, including self-employment and tem- porary jobs, driven by the digital platform economy (commonly known as the gig econ- omy), has thrust this form of work into the centre of social and political discussions. Among the European Union countries, Spain stands out with the highest volume of digital platform work. To explore the well-being effects of various gig economy employment arrangements, this study utilizes microdata from the Spanish Living Conditions Survey for 2018 and Google Trends data related to platforms like Deliveroo, Airbnb, Just Eat, Uber, and Freelance as a proxy for digital platform economy demand. Employing an economet- ric approach based on instrumental variables, the study reveals that the most detrimental well-being effects are observed among self-employed workers. Specifically, in terms of self-reported health, self-employed individuals (own-account workers) exhibit 125.8% low- er average health levels compared to permanent workers. These findings suggest that the heightened job insecurity and precariousness associated with self-employment outweigh the potential benefits arising from greater flexibility and autonomy in this type of work.
--	--

Keywords: digital platform economy, gig economy, digital platform work, self-employment, temporary jobs, well-being, self-reported health, happiness, life satisfaction.

JEL Classification: I31, J21, J81, J40.

[™]Corresponding author. E-mail: ainhoa.herrarte@uam.es

1. Introduction

The emergence of the digital platform economy, also known as the "gig economy", has been one of the most important transformations in the new world in the past decade (International Labour Office [ILO], 2018). Although the number of gig workers is still relatively small (1–3% of the global workforce according to Schwellnus et al. (2019)), it is estimated that digital platform work will continue to grow in the future and expand across sectors¹.

This economic revolution is causing a transformative and potentially severe impact on employment relationships, as it disrupts the general concept of "normal jobs". In the aftermath of the Great Recession (2008–2013), digital platforms have fundamentally reshaped the

¹ By country, the incidence varies from 9.4% in 2019 in United Kingdom, to 2.9 % in Germany or 0.8% in France or 0.5% to 1% of Norwegian working age adults (OECD/ILO/European Union, 2023, chapter 4).

Copyright © 2024 The Author(s). Published by Vilnius Gediminas Technical University

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/ licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

relationship between workers and companies. This transformation has given rise to what is commonly known as the gig or platform economy. As a result, labour markets recently exhibit a growing prevalence of non-traditional work arrangements (temporary jobs and self-employment). While non-traditional work arrangements have been the focus of attention for both scholars and the popular media over several decades, the swift emergence of platforms that utilize digital technologies to facilitate labour on a task-by-task basis has reignited interest in understanding how these work arrangements impact workers' well-being. Recent studies by Bajwa et al. (2018), Gross et al. (2018), Banks (2019), Berger et al. (2019), Apouey et al. (2020), Bérastégui (2021), and Escudero et al. (2023) delve into this critical area of inquiry. Insofar as these platforms create significant job opportunities, increase the just-in-time workforce, and provide temporal flexibility (De Stefano, 2016; Wood et al., 2018), such forms of employment might increase the well-being of workers. In fact, the demand for greater flexibility and a better work-life balance constitutes some of the drivers behind the growth of platform work (European Commission, 2017; OECD, 2019). However, as stated by Drahokoupil and Fabo (2018) and Kässi and Lehdonvirta (2016), these work arrangements may erode labour protection, be prone to poorer working conditions, and sustain economic insecurity and the unpredictability of working life. In such a case, lower levels of well-being may be observed among workers employed in this type of work arrangements.

Therefore, it is essential to understand how workers fare in these types of work arrangements associated to the gig economy (particularly temporary jobs and self-employment), which are becoming a prominent feature of 21st-century labour markets. The main purpose of this paper is to analyse the well-being of these workers in the Spanish labour market. In our study, we investigate the impact of digital platform-related work on various aspects of individual well-being, including health, happiness, and different facets of satisfaction (such as general satisfaction, financial satisfaction, satisfaction with work, personal relationships, and leisure time). To that end, we exploit microdata for the year 2018 from a specific module of the Spanish Living Conditions Survey included in European Union Statistics on Income and Living Conditions (EU-SILC).

We focus on Spain because it stands out among the EU countries with the highest volume of work through digital platforms. In particular, around 2.6% of the working population in Spain relies on digital platform work as their main form of employment (Pesole et al., 2018). This new labour model has placed the focus on the high job temporality and use of the "false self-employed" formula. As regards self-employment, some studies have pointed out that in many cases these work arrangements have ceased to be a free choice for workers in Spain (Monereo, 2016). Many of the new self-employed (freelancers), especially after the 2008 crisis, have found jobs through digital platforms, as indicated in a report by GoVup and the Spanish Digital Economy Association (GOVUP & Adigital, 2017). Moreover, according to the International Labour Organization (ILO, 2016), to the extent that jobs linked to digital platforms consist of project or task-based contracts and are of fixed-term duration or seasonal, temporary employment has become increasingly related to the digital platform economy. Since the beginning of the economic recovery in 2014, temporary hiring has become widespread in Spain, as noted by Felgueroso et al. (2017).

Our results indicate that self-employed workers have lower levels of individual well-being than other types of workers, especially in terms of self-reported health. Specifically, their average levels of self-reported health are 125.8% lower than that of permanent workers. In

contrast to the enhanced flexibility and autonomy associated with these atypical jobs prevalent in the digital platform economy, our findings indicate that factors such as employment insecurity, uncertainty, and social isolation have a more pronounced impact on workers' individual well-being.

The paper is organized as follows. Following the introduction, Section 2 reviews the most recent literature on the effects of the digital platform economy on the labour market. Sections 3 and 4 describe the datasets and provide a descriptive analysis of the types of work arrangements associated to the gig economy in Spain and its relation to individual well-being. Section 5 describes the empirical model and Section 6 discusses the results. Section 7 concludes.

2. Background: the digital platform economy and the labour market

The digital platform economy, or gig economy, is based on non-standard forms of employment (NSE, hereinafter) that are closer to "gigs" than traditional kinds of jobs (ILO, 2016). Over the past few decades, there has been a marked shift towards NSE in both industrialized and developing countries. Indeed, NSE has become a contemporary feature of labour markets around the world, as the use of such work arrangements has become more widespread across economic sectors and occupations. In 2013, NSE accounted for around a third of total employment in OECD countries (OECD, 2015). As stated in ILO (2015), NSE relies heavily on temporary and part-time positions filled by independent contractors and freelancers rather than full-time and permanent employees². Thus, the evolution of the digital platform economy is closely associated with these atypical forms of employment.

There is no accepted definition of the gig economy, but the literature has related it to non-standard work arrangements associated with both self-employment and temporary employment. With respect to the former, Abraham et al. (2019) utilize the taxonomy established in Abraham et al. (2018) to delineate a gig worker as an individual who is not remunerated with a wage or salary, lacks an implicit or explicit contract for ongoing employment, and operates without a predictable work schedule or earnings when working. Applying this definition to various alternative work arrangements, the authors classify independent contractors, freelancers, day labourers, and on-demand or platform workers as gig (related) workers. Thus, under this definition, gig workers should be part of the unincorporated self-employed. With respect to temporary employment, Valletta et al. (2020) defined gig jobs as short-term work arrangements based on contracted agreements, with a specific emphasis on tasks facilitated through online platforms that connect suppliers with demanders of products and services. In a similar vein, Glavin and Schieman (2022) underscored that temporary and short-term employment compensated on a piece-rate basis are key characteristics of the "gig economy". Based on an online survey, they distinguished between platform work (ride-hail and delivery services, remote online work, etc.), including dependent platform workers (their main job is app-based or web-based) and secondary platform workers, and non-platform workers (those reporting no platform work including temporary or permanent wage workers and traditional

² The classification of non-standard employment was the subject of discussion at the February 2015 ILO Meeting of Experts on Non-standard Forms of Employment (ILO, 2015).

self-employed). Finally, OECD/ILO/European Union (2023) offers a review of current sources and metrics for measuring digital platform employment, focusing exclusively on online and location-based services facilitated by digital labour platforms.

Based on all this literature, in our study, we contemplate temporary employment and self-employed being conscious that these two forms of employment are not fully identified with gig work, although they share the main features. Temporary employment encompasses various arrangements where workers are engaged for a specific duration. This includes fixed-term, project- or task-based contracts, seasonal or casual work, or day labour. For decades, the Spanish labour market has been characterized by a high level of temporary employment, especially compared to the EU average. This phenomenon has resulted in the creation of a dual labour market, where workers are employed under either fixed-term, temporary contracts, or regular, open-ended contracts. Although temporary employment decreased slightly during the years of the economic crisis³ due to the widespread destruction of temporary jobs, following the economic recovery temporary employment increased again to account for 22.7% of all jobs in 2018. In 2021, this figure was even higher, with over 24% of the Spanish workforce employed on temporary work contracts; a much higher share relative to other OECD countries⁴.

Self-employment has also been considered NSE in several studies on industrialized countries (see, for example, OECD, 2015). In the Spanish labour market, self-employment also increased over the period 2012–2016 and has stabilized at 15–16%⁵ in recent years; a slightly higher figure than the European average.

In principle, NSE can have both positive and negative aspects and hence affect workers in terms of their well-being. On the one hand, work arrangements associated to NSE might be valued and preferred by some workers who opt for this type of employment to achieve a better work-family balance and/or greater autonomy and sense of control in the case of self-employment. Nevertheless, NSE may be linked to precariousness and poorer working conditions. Non-standard workers often lack the same levels of employment protection, safeguards, and fringe benefits enjoyed by their counterparts in standard working arrangements. Due to this precariousness, workers in NSE may suffer greater levels of job insecurity⁶ – the subjective perception of losing one's job in the near future. During recessions or economic downturns, individuals often worry about the stability of their jobs, but they also may experience job insecurity due to organizational changes, modifications of the labour market regulation, mergers and restructuring plans, technological changes such as digitalization, etc. Although job insecurity is a problem that affects all individuals, specific workers are more exposed to greater risks, and the type of contract is one of the drivers of employment instability. In the metanalysis of Keim et al. (2014), the authors found that lower levels of

³ See Eurostat data (2000–2018): Temporary employees.

⁴ Employment – Temporary employment – OECD Data.

⁵ Around 10–11% for own-account workers.

⁶ Job insecurity is a complex phenomenon (see Greenhalgh & Rosenblatt, 2010, for a review of different definitions and measures). According to Ashford et al. (1989), job insecurity encompasses insecurity about losing the whole job (becoming unemployed), insecurity about losing specific job features (undesirable replacement, unpleasant tasks, privileges curtailed, etc.), and powerlessness to prevent the loss. A remarkable aspect of job insecurity is that it is a subjective perception, implying that different employees might perceive the same situation differently.

job insecurity are associated, among other factors, with having a permanent work, and in a more recent paper based on data from the European Working Conditions Surveys 2010 and 2015, Eichhorst and Tobsch (2017) found that, overall, fixed-term workers experience much lower levels of job security than those in permanent employment⁷. For the Spanish case, the influence of the type of contract on job stability and, hence, on the global job component of job insecurity is especially remarkable due to the severe segmentation of its labour market (Spain has one of the highest rates of fixed-term contracts within the EU). During the Great Recession (2008–2013), more than 3.4 million workers in Spain lost their jobs, of whom 61% were fixed-term workers. Hence, these factors associated with NSE may negatively affect individuals' well-being. In this respect, a recent meta-analysis concluded that employees in non-standard employment experience lower global subjective well-being (Fabrin-Petersen, 2022).

As regards temporary employment, a large body of research has shown evidence of the negative impact of this type of work arrangement on subjective well-being (Klein Hesselink & van Vuuren, 1999; Blanchard & Landier, 2002; Quesnel-Vallee et al., 2010; Robone et al., 2011; Carrieri et al., 2012; Dawson et al., 2017; Inanc, 2018; Karabchuk & Soboleva, 2020; Wu, 2023)⁸ and happiness (Scherer, 2009; Ponzo, 2011; Schumann & Kuchinke, 2020). However, some studies have reported a weak or non-negative impact of these work arrangements on workers' well-being (Sverke et al., 2000; Rodríguez, 2002; Bardasi & Francesconi, 2004; Silla et al., 2005; Cottini & Lucifora, 2013; Helbling & Kanji, 2018; Bartoll et al., 2019).

Research has also found that, overall, people who are self-employed exhibit consistently higher levels of subjective well-being measured either in terms of happiness or job satisfaction than those who are not self-employed (Blanchflower & Oswald, 1998; Blanchflower, 2000; Alesina et al., 2004; Andersson, 2008; Kawaguchi, 2008; Benz & Frey, 2008; Binder & Coad, 2013; Millan et al., 2013; Kara & Pretescu, 2018; Binder & Blankenberg, 2021). Binder and Blankenberg (2021) highlighted that while self-employed workers tend to report higher job satisfaction, this does not always translate into increased overall life satisfaction. This phenomenon is attributed to the diversity among self-employment arrangements, along with work characteristics, motivational factors, and institutional frameworks that differ across countries. In recent years, however, self-employed workers have become more heterogeneous. Although some individuals may experience satisfactory work quality and secure continuity of employment, for others, this particular work arrangement has become more commonly linked with terms such as "involuntary", "dependent", and "precarious" self-employment (Stone, 2006; Schulze et al., 2009; Kautonen et al., 2010; Westerveld, 2012; Eurofound, 2017).

Studies focusing specifically on gig workers and well-being outcomes are scarcer due

⁷ On the contrary, there is evidence suggesting that job insecurity plays a moderating rather than a mediating role between contract type and well-being. Several authors have observed that the association between job insecurity and psychological outcomes is more negative among permanent workers compared to temporary workers (De Cuyper & De Witte, 2006, 2007; Virtanen et al., 2002; Sverke et al., 2000). However, other authors have not found significant differences between temporary and permanent workers. For example, De Witte and Näswall (2003) regarding job satisfaction, or Dawson et al. (2015) and Russo and Terraneo (2020) regarding mental health.

⁸ For the Spanish case, Escudero et al. (2023) examined the relationship between precarious employment and psychological well-being (mental health). They concluded that, in comparison to short-term temporary contracts, self-employment, and atypical arrangements, the stability offered by permanent employment contracts contributes to greater well-being. However, this effect is observed only among workers who attain certain specific levels of job satisfaction beforehand.

mainly to the availability of accurate data to measure gig work (OECD/ILO/European Union, 2023). Among the few existing ones, Bajwa et al. (2018) emphasized that the health and well-being of gig workers are influenced by the design and operation of platform businesses. This includes several factors, such as worker classification, control over pricing and workflow, social isolation, and work-related stress arising from monitoring. In the same line, Gross et al. (2018) indicated that gig workers' mental well-being could be affected by precarity of finances, status, certainty and sociability. Bérastégui (2021) showed that gig work generates challenges for workers in terms of physical and social isolation, with the well-documented consequence of a lack of workplace social support, which directly affects workers' health and well-being. Apouey et al. (2020) studied the financial precarity and well-being outcomes of gig economy workers (especially drivers and bikers food delivery) in France before and during the mandatory lockdown of the COVID-19 crisis. They found that, although gig workers were more impacted by the crisis in terms of financial and physical health outcomes, drivers did not report higher levels of stress and anxiety compared to other precarious workers. Interestingly, bikers actually reported lower levels of stress and anxiety. In the framework of organizational theory, Keith et al. (2020) proposed a working model and highlighted that the gig economy could have both negative and positive effects on well-being, depending on the motivations for entering the gig economy and one's experience in the gig economy is likely to rely on the nature of the gig work - particularly how well the job resources balance out the demands. In this line, Berger et al. (2019) utilized administrative data from Uber along with a representative survey of London drivers. They discovered that drivers who prioritize flexibility as a key motivation for joining Uber tend to report elevated levels of subjective wellbeing. This underscores the significance of non-monetary factors in influencing the welfare of workers within the gig economy.

This paper attempts to contribute to this literature by providing recent empirical evidence on the subjective well-being effects of temporary and self-employment; two non-standard forms of employment that have been closely associated with the emergence of the digital platform or gig economy in recent years.

3. Data

To achieve the objectives of the paper, we exploit data from several sources that include information on self-employment and temporary employment. In a first stage, we provide an overview of the incidence and evolution of temporary and self-employment for Spain and the EU-15 for the period 2000–2018 using aggregated data from the European Union Labour Force Survey (EU-LFS). Specifically, we analyse the incidence of self-employment and temporary work on total employment, attending to several individual and labour characteristics such as gender, educational level, age, and occupation.

The analysis of the relationship between self-employment, temporary employment, and the individual well-being of workers is based on cross-sectional microdata from the 2018 wave of the Spanish Living Conditions Survey (ECV-2018); the Spanish sample of the EU-SILC.

The ECV-2018⁹ is well suited to achieve the purposes of this study as it includes a special module with self-reported information on various dimensions of subjective well-being: self-reported health, happiness, life satisfaction, satisfaction with financial situation, job satisfaction, satisfaction with personal relationships, and satisfaction with leisure time. To complement the ECV-2018 microdata and following the proposal of Apouey and Stabile (2019), we use Google search data on the amount of search activity related to the gig economy. In particular, we use Google trends data at the regional level (autonomous communities)¹⁰ on Deliveroo, Airbnb, Just Eat, Uber, and Freelance as a proxy of the digital platform economy demand to estimate the likelihood that an individual will be employed in a gig-type job (employee with a temporary contract or self-employed).

4. Descriptive analysis

4.1. Jobs related to the gig economy in Spain and in the EU-15

Our analysis starts with an overview of the occurrence and progression of self-employment and temporary employment in EU-15 during the period 2000-2018. We base our examination on aggregated data sourced from the European Union Labour Force Survey (EU-LFS). In general terms, the analysis reveals that self-employment and temporary employment rates are higher in Spain than the rest of the EU-15 countries, especially in terms of temporality. In particular, the percentage of self-employed persons in Spain was 14.9% in 2018, while the figure for the rest of the EU-15 was 13%,¹¹ while the temporary employment rate (measured over total employment) was 22.7% in Spain and 12.6% in the EU-15. As regards individual characteristics, self-employment tends to be more prevalent among men, whereas temporary employment is more common among women. Interestingly, the gap between women in Spain and the rest of Europe regarding temporary employment is widening¹². There is also an inverse relationship with educational level, although this is more evident in the case of temporary jobs. Additionally, older workers tend to engage in self-employment more frequently, whereas temporary employment is predominantly concentrated among young people. As regards self-employment, Figure 1 shows that the incidence of this type of work is lower among workers with tertiary education in Spain than the European average, especially among women. This finding indicates that tasks and occupations linked to these jobs in other European countries tend to require a higher level of qualification (such as liberal professions). Additionally, the data suggest that European women with higher education often choose more flexible work arrangements that prioritize work-family balance more than their Spanish counterparts. Both in Spain and in Europe, the low frequency of self-employment among young people is noteworthy.

⁹ The ECV is conducted on a sample of around 13,000 households and involves approximately 35,000 individuals. Its objective is to collect comparable data on income, poverty, social exclusion, and living conditions across time. Around 90% of the collected data comprises annual variables. The rest are either modules that are collected every three or six years or modules conducted ad-hoc to respond to policy needs.

¹⁰ NUTS-2.

¹¹ In terms of self-employed persons without employees (own-account workers) these figures were 10.4% for Spain and 9.1% for the rest of the EU-15.

¹² See Figures A1 and A2 in the Appendix.

Figure 2 presents data on temporary employment. Notably, Spain exhibits a higher prevalence of temporary employment compared to the other EU-15 countries across all demographic groups. Furthermore, women experience a significantly higher incidence of temporary work. A distinctive feature in Spain is that, unlike men, primary and secondary educated women demonstrate similar rates of temporary employment. These figures are only lower



Figure 1. Self-employment in 2018 by gender, educational level, and age (% over total employment) (source: own elaboration based on data from the EU Labour Force Survey (Eurostat, 2018))



Spain (2018) Rest of UE-15 (2018)

Figure 2. Temporary employment by gender, educational level, and age in 2018 (% over total employment) (source: own elaboration based on data from the EU Labour Force Survey (Eurostat, 2018))

for women with tertiary education. In contrast, the association between job temporality and education is consistent for both men and women across the rest of Europe. These findings suggest that, unlike the broader EU-15 context, Spanish women require more years of education than men do to attain more stable employment. Lastly, there is a significant disparity between Spain and the rest of the EU-15 (26 percentage points) concerning the incidence of temporary work among the youngest workers.

4.2. Employment status and individual subjective well-being

Our analysis of the effects of the digital platform economy or gig economy on individual subjective well-being is based on two subsamples of individuals aged 16-64 years drawn from the ECV-2018 microdata. The first subsample includes 11,377 employed individuals (employees with permanent contracts, employees with temporary contracts, and the self-employed).¹³ The second subsample also considers unemployed and inactive individuals, amounting to a total of 16,527 individuals. Using the first sample we will compare self-employed and temporary workers in terms of well-being, specifically contrasting them with permanent workers. The second sample will enable us to conduct a comparative analysis not only with respect to self-employed and temporary workers but also concerning unemployed and inactive workers. By examining these distinct groups, we aim to gain insights into the well-being implications of different employment arrangements. According to ILO definition, self-employment comprises both self-employed with employees and own-account workers. Self-employed with employees constitute employers as they are people whose primary activity is self-employment and who employ others (and have the authority to decide about how to run the business). They usually represent entrepreneurs and are not considered as atypical forms of employment. In contrast, own-account workers are self-employed working on their own, such as contractors and freelancers, more associated with the type of non-standard employment common in the gig or digital platform economy. Hence, it should be remarked that, similarly to other studies (OECD, 2015; Abraham et al., 2019; Valletta et al., 2020; Bérastégui, 2021), self-employment in the empirical analysis refers exclusively to own-account workers, excluding self-employed persons with employees (employers).

Across the entire sample, we observe that 45.8% of individuals fall into the category of permanent workers, 15.5% are unemployed, and 15.7% of individuals report to be inactive. Approximately 23% of the sample is engaged in one of the typical forms of gig economy work, which includes self-employment (7.7%) and temporary employment (15.3%)¹⁴. These gig economy forms of employment constitute 33.5% of employed individuals in the sample, with temporary employees accounting for 22.2% and the self-employed comprising 11.2%. Figure 3 provides insights into self-employment and temporary employment across Spanish regions (autonomous communities) and distinguishes between white-collar (managerial, professional, technical, or administrative positions) and blue-collar (skilled workers in service, manufacturing, or construction sectors, operators, and workers in elementary occupations). Incidence of gig economy jobs is much lower in white-collar occupations across all regions,

¹³ See Table A1 in Appendix A for a detailed description of the different employment status in the ECV.

¹⁴ Table A2 in the Appendix displays the sample characteristics of both samples.



Notes: White-collar occupations (Directors and managers, Scientific technicians and professionals, Support technicians, Accountants and administrative staff). Blue-collar occupations (Service workers and salespersons, Skilled agricultural workers, Skilled artisans and skilled industrial and construction workers, Operators and assemblers, Elementary occupations). Armed forces occupations have not been included. Self-employed refers to own-account workers.



being more prevalent in blue-collar occupations. Notably, Extremadura exhibits a 58% incidence of self-employment and temporary employment in blue-collar occupations. Similar patterns are observed in Castilla-La Mancha, Andalusia, and Murcia. In contrast, the Basque Country shows the smallest differences between blue-collar (36%) and white-collar (34%) occupations.

Table 1 delves into the interplay between gig economy jobs and individual well-being. It offers a descriptive analysis of various dimensions of subjective well-being, based on individuals' employment status. For all dimensions, this is self-reported information. Here are the specifics. In the case of satisfaction variables, responses are recorded on a scale of 1 to 11, where value 1 signifies "not at all satisfied," and value 11 represents "completely satisfied." In the case of self-reported health and happiness, the variables use a scale ranging from 1 (very bad) to 5 (very good). This self-reported information sheds light on how different employment arrangements impact well-being¹⁵. Numerous studies have demonstrated that subjective well-being measures possess strong predictive power for relevant phenomena. These measures are correlated with a variety of observable indicators, including physical health and longevity (Danner et al., 2001), suicide rates, macroeconomic fluctuations (Di Tella et al., 2003), and unemployment (Clark et al., 2008), among other factors. These self-reported measures also show reasonable consistency, as they correlate well with each other and with

¹⁵ See Table A3 in Appendix A for a detailed description of the well-being variables.

	Permanent contracts			Temporary contracts		Self-employed (own-account workers)		loyed	Inactive	
	Mean	SE	Mean	SE	Mean	SE	Mean	SE	Mean	SE
Self-reported Health	4.12	0.65	4.15	0.68	4.04	0.68	3.88	0.81	3.60	0.99
Happiness	4.02	0.86	3.97	0.89	3.91	0.87	3.65	1.01	3.81	0.98
Life satisfaction	8.80	1.38	8.43	1.62	8.59	1.50	7.48	2.06	8.16	1.95
Financial satisfaction	7.91	1.74	7.22	1.99	7.60	1.77	5.72	2.43	7.18	2.19
Personal relationship satisfaction	9.32	1.38	9.28	1.39	9.29	1.33	9.08	1.61	9.18	1.59
Job satisfaction	8.41	1.72	8.05	1.93	8.26	1.85	-	-	-	-
Leisure satisfaction	7.42	2.18	7.41	2.18	6.88	2.50	8.02	2.17	8.33	2.08
N	7,57	70	2,528		1,279		2,560		2,590	

Table 1. Main descriptives of well-being variables	Table	1.	Main	descrip	ptives	of	well-being	variables
--	-------	----	------	---------	--------	----	------------	-----------

Note: Almost all pairwise mean differences across economic status are statistically different from zero. Test results are available upon request.

alternative measures based on information provided by family and friends and a wide range of psychological and psychosocial indicators (Cacioppo & Patrick, 2008).

As Table 1 shows, employed workers report higher levels of individual well-being than unemployed or inactive individuals except in leisure satisfaction. In general terms, well-being seems to be higher for employed workers with permanent contracts, especially in the dimension of financial satisfaction. Employed workers with a permanent contract report a mean score of 7.91 in financial satisfaction, which is 10% higher than the score reported by fixedterm employees and 4% higher than self-employed workers. To a lesser extent, permanent employees report a mean value of job satisfaction and life satisfaction that is 4% and 2% higher than fixed-term employees and self-employed workers, respectively. In contrast, no significant differences were found between self-employed and temporary workers in the mean values of the different well-being indicators, with the only exception being leisure satisfaction. Both permanent and fixed-term workers report much higher levels of satisfaction with leisure time than self-employed workers (around 8% higher).

These data suggest that the type of contract exerts an effect on individual well-being. In the next section we aim to shed more light on the impact of these atypical working arrangements on individual well-being.

5. Empirical model and econometric strategy

To examine how self-employment and temporary work affect individual well-being, we utilize the following specification:

$$SWB_{ii} = \alpha + \beta LMS_{ii} + \gamma X_{ii} + \delta Y_i + \epsilon_{ii}, \qquad (1)$$

where SWB_{ij} represents the subjective well-being of individual *i* who lives in region *j*. As mentioned, we consider seven dimensions of well-being: self-reported health, happiness, life

satisfaction, satisfaction with financial situation, job satisfaction, satisfaction with personal relationships, and satisfaction with leisure time. For the estimation of Equation (1), the different SWB measures are considered to encompass a cardinality-type classification. While the assumption of cardinality rather than ordinality is not relevant for the results (Ferrer-i-Carbonell & Frijters, 2004), it has the advantage of producing coefficients that can be directly interpreted as marginal effects.

LMS_{ij} is a vector that includes several dummy variables related to the employment situation of individual *i* in region *j*. Specifically, for the analysis of the subsample of employed persons, the vector LMS_{ij} includes the dummy variable *GIG_Emp* (which takes the value of 1 if the person is self-employed or is on a temporary contract). In addition, for the analysis of the total sample of individuals (employed, unemployed, and inactive) the vector LMS_{ij} includes the dummy variables *Unemployed* and *Inactive*, which take the value of 1 if the individual is unemployed or inactive, respectively. For both samples, the reference category in the estimations is *Permanent* (being employed with a permanent contract). Moreover, to explore in greater depth which types of contractual arrangements closely linked to the gig economy may lead to lower levels of well-being, we consider the self-employed (*Self-Employed*) and temporary employees (*Temporary_Employed*) separately.

Vector X_{ij} includes several socio-demographic characteristics at the individual and household levels. The individual characteristics include gender, age, marital status, and level of education, and the household characteristics are total household size, number of children, household income, and an indicator of whether the household has difficulties in making ends meet. This vector also includes a dummy variable that distinguishes between blue-collar¹⁶ and white-collar occupations. Finally, vector Y_j includes the average income and unemployment rate at regional level.

Our empirical strategy starts with the estimation of equation (1) by ordinary least squares (OLS) for both subsamples (employed individuals and total sample). In the first model (OLS-1) we have jointly considered self-employed and temporary workers under the category *GIG_Emp*, while the second model (OLS-2) considers both types of employment arrangements separately in the estimations (*Self_employed* and *Temporary_Emp*).

An inherent problem in determining the sign of the causal effect that the type of work most closely related to the digital platform economy may have on individual well-being is the possibility of a reverse causality and selection problem. In other words, levels of individual well-being might affect the probability that an individual is self-employed or a temporary employee. Moreover, other unobservable factors may affect both the likelihood of being employed in a gig-economy-related job and individual well-being. To address these problems, and in line with previous work (Apouey & Stabile, 2019; Berger et al., 2019), Equation (1) is also estimated using an instrumental variables methodology (IV).

Following Apouey and Stabile (2019), we use Google Trends data at the regional level (autonomous communities)¹⁷ for the year 2018 corresponding to the number of Google

¹⁶ Based on major groups 5 to 9 of the ISCO-2008 (Service workers and salespersons, Skilled agricultural workers, Skilled artisans and skilled industrial and construction workers, Operators and assemblers, Elementary occupations). Armed forces are excluded from the analysis.

¹⁷ Autonomous communities (NUTS-2) is the highest level of regional disaggregation provided by Google trends data for Spain.

searches for different digital platforms related to the gig economy (Deliveroo, Airbnb, Just Eat, Uber, and Freelance). We use these Google trends data as a proxy variable for the gig economy demand and hence as an instrument to estimate the probability that an individual is employed in one of the two contractual arrangements most closely related to the gig economy (self-employment or temporary job). These web search data are increasingly being used as measures of economic activity or indicators of the demand in this type of digital platform economy, which also make it possible to obtain predictions on the evolution of unemployment among other macroeconomic variables (see, e.g., D'Amuri & Marcucci, 2017).

More precisely, using a sample of searches, Google Trends provides the percentage of a region's searches for a given word divided by the percentage of searches on a given word in that region with the highest share of searches for that same word multiplied by 100. The resulting data is therefore relative to the region with the highest share of searches at time t equal to 100.

Specifically, for region j at time t the score for the word "W" is defined as follows:



Notice that our identification strategy relies on the assumption that Google searches for different digital platforms do not exert a direct impact on individual well-being, except through the employment situation. Nonetheless, these Google searches might also be a proxy for the total economic activity of the region, which may be correlated with individual levels of well-being. To address this issue, we have included in vector Y_j of equation (1) the average income and unemployment rate at the regional level.

6. Results

Table 2 and Table 3 display the results from the OLS and IV regressions for the sample of employed individuals and the full sample, respectively. Since we find evidence of endogeneity in all the estimates¹⁸, not controlling for the reverse causality problem (OLS-1 and OLS-2) would result in misleading conclusions. Hence, our comments will be based on models IV-1 and IV-2 where we control for possible sources of endogeneity and self-selection.

Starting with the sample of employed individuals (Table 2), the results indicate that jointly considering the types of jobs most closely linked to the digital platform economy affects negatively and significantly almost all the dimensions of individual well-being (Table 2, Model IV-1). The largest and most highly significant effect appears in the dimension of life satisfaction (the coefficient is –2.994), followed by self-reported health. The lowest effect occurs in the leisure dimension. Additionally, when considering self-employment and temporary employment separately (Table 2, Model IV-2), we find that the negative impact on well-being is mostly caused by self-employment. This is an interesting issue, since when we do not

¹⁸ The endogeneity test of endogenous regressors is significant at the 1% level in all estimates.

		Self-repor	ted health			Нарр	oiness	
	OLS-1 ^(a)	OLS-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	OLS-1 ^(a)	OLS-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)
	-0.036*		-2.326***	-	-0.074***	-	-2.011***	-
GIG_Emp	[0.017]		[0.506]		[0.018]		[0.483]	
	-	-0.027	-	-5.182***	-	-0.075**	-	-2.899**
Self_employed		[0.024]		[1.337]		[0.027]		[0.941]
Temporary_	-	-0.041*	-	-0.621	-	-0.074***	-	-0.059
Emp		[0.020]		[0.679]		[0.022]		[0.478]
	Life s	atisfaction	·			Financial s	atisfaction	<u>.</u>
CIC Emp	-0.113***	-	-2.994***	-	-0.101***	-	-1.798***	-
GIG_Emp	[0.019]		[0.608]		[0.017]		[0.426]	
Solf omployed	-	-0.038	-	-6.870***	-	-0.023	-	-5.042***
Self_employed		[0.027]		[1.773]		[0.024]		[1.408]
Temporary_	-	-0.157***	-	0.278	-	-0.147***	-	0.859
Emp		[0.022]		[0.901]		[0.020]		[0.715]
Pers	onal relation	onships sa	tisfaction			Leisure sa	atisfaction	
CIC Emp	-0.031	-	-1.376**	-	-0.060**	-	-1.353**	-
GIG_Emp	[0.020]		[0.436]		[0.020]		[0.434]	
Self_employed	-	0.005		-4.448**	-	-0.166***	-	-3.035**
Sell_ellipioyed		[0.028]		[1.441]		[0.031]		[0.982]
Temporary_	-	-0.052*		1.671*	-	0.003	-	-0.858+
Emp		[0.023]		[0.732]		[0.023]		[0.499]
	Job s	atisfaction						
GIG_Emp	-0.079***	-	-1.396**	-				
GIO_EIIIp	[0.020]		[0.441]					
Self_employed	-	0.006	-	-3.787**				
		[0.030]		[1.155]				
Temporary_	-	-0.129***	-	-0.065				
Emp		[0.024]		[0.587]				
N	11,377	11,377	11,377	11,377	11,377	11,377	11,377	11,377

 Table 2. Effects of the work-related with the digital platform economy on individual well-being (employed workers). Main results

Standard errors in brackets. + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001.

(a) Variable *Gig* considers jointly self-employment and temporary employment; (b) Self-employment and temporary employment appear separately. Self-employed refers to own-account workers.

Controls: gender, age, educational level, marital status, number of children, household size, household income, household with difficulties to make ends meet, white/blue collar, regional unemployment rate, regional mean income.

Instruments: We use data on Just Eat, Uber, and Airbnb (Google trends) for Model IV-1. Just Eat, Uber, Airbnb, and population size for IV-2. Alternative estimates were made with different combinations including the number of Deliveroo and Freelance searches and finally the number of Just Eat, Uber, and Airbnb searches was chosen as the best instruments. Results not included but available upon request.

control for the reverse causality problem (OLS-2), we obtain the false result that temporary jobs are associated with the lowest levels of well-being, while the self-employed show only slightly lower levels of well-being than permanent employees. However, a completely different picture emerges when a methodological approach that allows controlling for the reverse causality problem is applied. In this case, we find that self-employed workers display the worst results in practically all the well-being dimensions analysed (Model IV-2), while temporary workers do not exhibit significantly lower levels of well-being than their counterparts in permanent jobs. However, it is worth noticing that being employed in a temporary job implies a higher individual well-being in the dimension of satisfaction with personal relationships¹⁹.

As regards the size of the effect, in Figure 4, we plot the differences (in %) in all dimensions of well-being between types of employment related to the gig economy and the corresponding levels among permanent workers (reference category). We sort the dimensions of well-being from the smallest to the largest effect. We find that gig-economy-related workers display average levels of self-reported health that are 56.5% lower than permanent workers.



Notes: The difference in well-being is computed by dividing the corresponding coefficients for gig-related employment, self-employed and temporary employees from the estimations of models IV-1 and IV-2 (Table 2, sample of employed workers) by the mean value of each well-being indicator for permanent employees (reference category; see column 2 in Table 1). For instance, the –56.5% is the result of dividing the coefficient of gig related employment in the self-reported health dimension (Model IV-1) by the mean value of self-reported health among permanent employees (–2.326/4.12). Coefficients of temporary employment are only statistically significant for the dimension of personal relationships (see Table 2, Model IV-2). Self-employed refers to own-account workers.

Figure 4. Differences in well-being for self-employed and temporary workers vs. permanent employees (%) (source: own elaboration (ECV-2018 and Just Eat, Uber, and Airbnb data from Google trends))

¹⁹ We have performed some robustness checks. Following a referee's comment, we have compared IV estimates distinguishing by full-time and part-time employment. We have also considered a more restricted definition of self and temporary workers excluding those who declare not having access to an internet connection at home, and a second restricted definition only considering self and temporary workers in the service sector (and have an internet connection). The main results hold. See Appendix B (Tables B1–B2) for a detailed explanation of the robustness analysis.

This percentage is 125.8% in the case of the self-employed. Regarding other dimensions of well-being, the adverse impact of non-standard work arrangements commonly linked to the gig economy is less pronounced. However, even in the dimension with the lowest negative effect (job satisfaction), self-employees exhibit significantly lower levels of well-being (36.1% lower) than individuals in permanent jobs. Therefore, our results provide evidence that the negative impact of non-standard work arrangements most commonly associated with the gig economy on individual well-being is mainly driven by self-employment. This is a novel result since most of the literature has found that greater flexibility and autonomy in self-employment usually has a positive effect on well-being (Wang et al., 2022; Vučeković et al., 2023).

The negative relationship between the non-standard work arrangements most related with the gig economy and individual levels of well-being still holds. Furthermore, this effect is slightly amplified when the analysis includes unemployed and inactive individuals (Table 3). The most detrimental well-being effects are observed again in the dimensions of life satisfaction, followed by happiness and self-reported health (coefficients in Model IV-1). When we consider different types of labour market status, we find that self-employed workers again suffer the largest negative effect with respect to permanent employees, followed by unemployed and inactive individuals (Model IV-2 in Table 3). In contrast, temporary employment does not have a significant negative effect on individual well-being, and its positive influence on the dimension of satisfaction with personal relationships remains.

The negative impact of self-employment on well-being can partly be attributed to the circumstances during the Great Recession (starting in 2008). Many workers compelled to opt for this form of non-standard employment as the sole means to avoid unemployment, even though it might not have been their preferred choice. Consequently, the heightened insecurity and precariousness associated with necessary self-employment outweighed the potential benefits arising from the greater flexibility and autonomy inherent in this type of work²⁰. Interestingly, related research has also found a positive effect of self-employment on individual well-being, primarily driven by midsize and large employers. This underscores the nuanced impact of different employment arrangements on our overall sense of well-being.

This type of self-employee usually displays specific characteristics, such as high levels of motivation or more ability to recognize opportunities among others, which used to be positively related to individual well-being. In contrast, farmers, dependent freelancers, and own account workers generally have fewer of these features and tend to have lower levels of mental well-being (Gevaert et al., 2018). In a similar vein, Figure 5 shows the differences in dimensions of well-being for the self-employed, temporary workers, the unemployed, and inactive individuals with respect to permanent employees (reference category). The smallest difference is observed in the dimension of leisure satisfaction. In this well-being dimension, being self-employed leads to a 64.8% lower average satisfaction levels. In contrast, the largest difference appears in self-reported health, where the well-being of self-employed workers is, on average, 166.3% lower than that of permanent employees. The detrimental effects of being unemployed or inactive on well-being range from 1% (personal relationships) to 30.4%

²⁰ Binder and Blankenber (2021) emphasized that individuals who are self-employed out of necessity often exhibit lower levels of autonomy in their respective occupations, display diminished intrinsic motivation, and may also lack the personality traits typically associated with an entrepreneurial personality.

Table 3. Effects of the work-related with the digital platform economy on individual we	l-being
(Total sample). Main results	

		Self-repor	ted health			Нарр	oiness			
	OLS-1 ^(a)	OLS-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	OLS-1 ^(a)	OLS-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)		
CIC Emp	-0.051**	-	-3.439***	-	-0.070***	-	-3.624***	-		
GIG_Emp	[0.016]		[0.814]		[0.018]		[0.862]			
Salf amployed	-	-0.021	-	-6.850***	-	-0.076**	-	-5.492***		
Self_employed		[0.024]		[1.837]		[0.026]		[1.577]		
Temporary_	-	-0.067***	-	-0.785	-	-0.067**	-	-0.262		
Emp		[0.019]		[0.962]		[0.021]		[0.826]		
Unomployed	-0.171***	-0.172***	-1.608***	-1.251***	-0.232***	-0.232***	-1.740***	-0.976**		
Unemployed	[0.021] [0.021] [0.347]		[0.360]	[0.023]	[0.023]	[0.368]	[0.309]			
Inactivo	-0.415***	-0.415***	-1.701***	-1.446***	-0.108***	-0.108***	-1.457***	-0.829**		
Inactive	[0.023]	[0.023]	[0.311]	[0.327]	[0.023]	[0.023]	[0.329]	[0.281]		
	Life s	atisfaction	^		Financial satisfaction					
CIC Emp	-0.117***	-	-5.167***	-	-0.110***	-	-2.786***	-		
GIG_Emp	[0.018]		[1.130]		[0.016]		[0.688]			
Salf amployed	-	-0.041	-	–11.116***	-	-0.041+	-	-6.819***		
Self_employed		[0.026]		[2.940]		[0.023]		[1.921]		
Temporary_	-	-0.159***	-	0.752	-	-0.149***	-	0.861		
Emp		[0.022]		[1.539]		[0.019]		[1.006]		
Unomployed	-0.459***	-0.462***	-2.602***	-1.594**	-0.496***	-0.499***	-1.631***	-1.057**		
Unemployed	[0.024]	[0.024]	[0.482]	[0.576]	[0.021]	[0.021]	[0.293]	[0.377]		
Inactive	-0.178***	-0.179***	-2.095***	-1.320*	-0.103***	-0.105***	-1.119***	-0.687*		
mactive	[0.024]	[0.024]	[0.432]	[0.524]	[0.020]	[0.020]	[0.263]	[0.342]		
Pers	sonal relati	onships sa	tisfaction			Leisure sa	atisfaction			
CIC Emp	-0.027	-	-2.345***	-	-0.058**	-	-1.958**	-		
GIG_Emp	[0.019]		[0.690]		[0.020]		[0.634]			
Salf amployed	-	0.011	-	-6.647**	-	-0.181***	-	-4.810***		
Self_employed		[0.028]		[2.193]		[0.031]		[1.438]		
Temporary_	-	-0.048*	-	2.693*	-	0.01	_	-0.655		
Emp		[0.022]		[1.148]		[0.022]		[0.753]		
Inomployed	-0.064**	-0.065**	-1.048***	-0.093	0.394***	0.399***	-0.412	-0.385		
Unemployed	[0.024]	[0.024]	[0.294]	[0.430]	[0.024]	[0.024]	[0.270]	[0.282]		
Inactivo	-0.028	-0.029	-0.908***	-0.154	0.449***	0.451***	-0.272	-0.293		
Inactive	[0.024]	[0.024]	[0.264]	[0.391]	[0.024]	[0.024]	[0.242]	[0.256]		
N	16,527	16,527	16,527	16,527	16,527	16,527	16,527	16,527		

Note: Standard errors in brackets. + *p* < 0.1, * *p* < 0.05, ** *p* < 0.01, *** *p* < 0.001.

(a) Variable *Gig* considers jointly self-employment and temporary employment; (b) Self-employment and temporary employment appear separately. Self-employed refers to own-account workers.

Controls: gender, age, educational level, marital status, number of children, household size, household income, household with difficulties to make ends meet, white/blue collar, regional unemployment rate, regional mean income.

Instruments: We use data on Just Eat, Uber, and Airbnb (Google trends) for Model IV-1. Just Eat, Uber, Airbnb, and population size for Model IV-2. Alternative estimates were made with different combinations including the number of Deliveroo and Freelance searches and finally the number of Just Eat, Uber, and Airbnb searches was chosen as the best instruments. Results not included but available upon request.

(self-reported health) in the case of unemployed workers, and from 1.7% (personal relationships) to 35.1% (self-reported health) in the case of inactive individuals. Although not significantly different from zero in almost all the well-being dimensions, temporary employees seem to experience higher levels of well-being in the dimension of personal relationships by 28.9% with respect to permanent workers.

Overall, our results highlight the importance of correcting for the endogeneity of self-employment and temporary employment when analysing the influence of these types of jobs on individual well-being. To check the validity of the instruments, Table 4 displays the underidentification and overidentification tests, Anderson canon correlation LM statistic, and Sargan statistic using as instruments the number of Google searches for "Just Eat", "Uber" and "Airbnb" (for Model IV-1) plus the regional²¹ population size for Model IV-2. In almost all estimates, both tests lead us to reject the hypothesis of weak instruments, hence confirming the validity of the Google Trends data linked to the digital platform economy as instruments.

In sum, our results seem to indicate that the greater job insecurity and precariousness experienced by self-employees (own-account workers) exceed the potential favourable effects related to the greater flexibility and autonomy of these types of workers. Thus, our findings align with prior research in the literature, which indicates that the experience of job loss and the stress associated with factors such as low support (Syrett, 2016) or high workloads and



Notes: The differences in well-being is computed by dividing the corresponding coefficients for selfemployed, temporary employment, unemployment, and inactivity from estimations for Model IV-2 (Table 3, total sample) by the mean value of each well-being indicator for permanent employees (reference category; see second column in Table 1). Coefficients of temporary employment are only statistically significant in the personal relationships dimension (Table 3, Model IV-2). Self-employed refers to own-account workers.

Figure 5. Differences in well-being for self-employed, temporary workers, unemployed and inactive vs. permanent employees (%) (source: own elaboration (ECV-2018 and Just Eat, Uber, and Airbnb data from Google trends))

²¹ Spanish Autonomous Communities (NUTS2). Figures referred to 2018.

long working hours (Hyytinen & Ruuskanen, 2007; Nordenmark et al., 2012) in gig economy roles lead to overall unfavourable outcomes (De Witte, 1999, 2005; Sverke et al., 2002; Cheng & Chan, 2008; Probst, 2008).This effect is particularly pronounced for the group of self-employed individuals we analysed in this study – specifically, freelancers or independent professionals. Empirical evidence suggests that this group faces the poorest working conditions and mental well-being. Several factors may contribute to this situation: over-commitment tendencies (Syrett, 2016); low autonomy, economic dependence, and financial hardship (Böheim & Mühlberger, 2009); or the phenomenon of Effort-Reward Imbalance (Ertel et al., 2005). These combined factors underscore the unique challenges faced by self-employed individuals in terms of their well-being.

	Er	nploye	d workers			Total s	ample	
	Sarg	gan stat	tistic (overi	dentific	ation test c	of all ins	struments):	
	IV-1 ^(a)		IV-2 ^(b)		IV-1 ^(a)		IV-2 ^(b)	
Self-reported health	13.864	***	1.622		6.764	*	0.284	
Happiness	8.072	*	22.828	***	4.295		12.605	**
Life satisfaction	16.694	***	4.975	+	18.515	***	7.037	*
Financial satisfaction	10.832	**	0.506		10.217	**	0.806	
Job satisfaction	19.279	***	4.769	+	-		-	
Personal relationships satisf.	7.692	*	4.094		6.764	*	6.641	*
Leisure satisfaction	11.892	**	5.1	+	12.3	**	1.87	
	Unde	eridenti	fication tes	t (Ande	rson canor	n. corr. l	M statistic)
	IV-1 (a)		IV-2 (b)		IV-1 (a)		IV-2 (b)	
	32.583	***	16.453	***	24.453	***	15.432	**

 Table 4. Underidentification test and overidentification test of all instruments in Instrumental variables

 estimates

Note: + *p* < 0.1, * *p* < 0.05, ** *p* < 0.01, *** *p* < 0.001.

7. Conclusions

In this study, we examined the impact of being employed in job types closely associated with the digital platform or gig economy on various dimensions of individual well-being in Spain for the year 2018. Spain is among the EU countries with the highest volume of digital platform work. The findings indeed validate a greater prevalence of self-employment and, particularly, temporary employment in Spain when compared to the other EU-15 countries over the period spanning from 2000 to 2018. According to our data, the self-employed have lower levels of individual well-being than the rest of workers, with the greatest negative impact on self-reported health (125.8% lower than permanent workers). Our results also underline the relevance of correcting for the endogeneity of self-employment and temporary employment. Once we control for the reverse causality problem using Google trends data as instruments of digital platform economy demand, temporary workers do not exhibit significantly lower

levels of well-being than their counterparts in permanent jobs. Thus, we can conclude that the negative impact on well-being is mostly caused by self-employment (own-account workers).

Our estimates indicate that the adverse impact on well-being, stemming from factors like job insecurity and precariousness associated with self-employment, outweighs the potential positive impact arising from the greater flexibility and autonomy inherent in this type of work. However, in our estimations we do not find evidence that temporary employment produces negative effects on individual well-being. These findings indicate that the profile of self-employed workers has changed since the Great Recession of 2008–2013, as now-adays self-employment is no longer a voluntary decision between being an employee or self-employed but rather the result of the only opportunity for employment in a strongly deteriorated labour market.

Some limitations should be mentioned. To date, there is no standard measure that allows determining the true incidence of the digital platform economy in society because existing data sources have problems capturing workers engaged in "gig jobs". In this sense, the only relevant progress is the Online Labour Index²² which measures the use of digital platforms in real time for all countries and occupations.

Furthermore, it is important to note that while self-employment and temporary employment are closely associated with the digital platform economy, they do not provide a fully accurate representation of gig jobs. Consequently, the findings of this study should be interpreted with caution. To gain a more precise understanding of the impact of the digital platform economy on society, it would be essential to collect higher quality data specifically focused on professionals working within and for digital platforms. This could be achieved, for example, by means of a register kept by the platforms themselves that would make it possible to obtain a census with detailed information on these workers. In addition, it would be very useful to be able to monitor workers in this type of non-standard jobs (e.g., through longitudinal databases) in order to correct the usual potential biases arising from both the self-selection of individuals in these jobs and the existence of unobservable factors that may simultaneously affect employment decisions and other types of individual decisions or situations; topics of study that could be of interest to researchers.

Funding

The authors acknowledge the financial support from the Fundación Bancaria "la Caixa" under Grant of the Observatorio Social "la Caixa" to support social research projects: vocational training, school drop-outs, and job insecurity; the Government of Spain under Grants PID2019-111765GB-I00 and PID2022-137819NB-I00; and the Regional Government of Andalusia under Grant PY18-4115.

Author contributions

These authors contributed equally to this work.

²² See Kässi and Lehdonvirta (2018).

Disclosure statement

All authors certify that they have no affiliations with or involvement in any organization or entity with any financial interest or non-financial interest in the subject matter or materials discussed in this manuscript.

References

- Abraham, K. G., Haltiwanger, J., Sandusky, K., & Spletzer, J. (2018). Measuring the gig economy: Current knowledge and open issues (NBER Working Paper 24950). National Bureau of Economic Research. https://doi.org/10.3386/w24950
- Abraham, K. G., Haltiwanger, J., Sandusky, K., & Spletzer, J. (2019). The rise of the gig economy: Fact or fiction? AEA Papers and Proceedings, 109, 357–361. https://doi.org/10.1257/pandp.20191039
- Alesina, A., Di Tella, R., & MacCulloch, R. (2004). Inequality and happiness: Are Europeans and Americans different? Journal of Public Economics, 88(9), 2009–2042. https://doi.org/10.1016/j.jpubeco.2003.07.006
- Andersson, P. (2008). Happiness and health: Well-being among the self-employed. The Journal of Socio-Economics, 37(1), 213–236. https://doi.org/10.1016/j.socec.2007.03.003
- Apouey, B., & Stabile, M. (2019). The effects of self and temporary employment on mental health: The role of the gig economy in the UK (Working Paper 2019/23/EPS). INSEAD The Business School of the World.
- Apouey, B., Roulet, A., Solal. I. & Stabile, M. (2020). Gig workers during the COVID-19 crisis in France: Financial precarity and mental well-being. *Journal of Urban Health*, 97(6), 776–795. https://doi.org/10.1007/s11524-020-00480-4
- Ashford, S. J., Lee, C., & Bobko, P. (1989). Content, causes, and consequences of job insecurity: A theorybased measure and substantive test. Academy of Management Journal, 32(4), 803–829.
- Bajwa, U., Gastaldo, D., Di Ruggiero, E., & Knorr, L. (2018). The health of workers in the global gig economy. *Global Health*, 14, Article 124. https://doi.org/10.1186/s12992-018-0444-8
- Banks, C. (2019). Health, safety, well-being and economic security implications of gig work: An interdisciplinary perspective. "Think Piece" for ILO World Day for Safety and Health at Work. Interdisciplinary Center for Healthy Workplaces, UC Berkeley.
- Bardasi, E., & Francesconi, M. (2004). The impact of atypical employment on individual wellbeing: Evidence from a panel of British workers. Social Science and Medicine, 58(9), 1671–1688. https://doi.org/10.1016/S0277-9536(03)00400-3
- Bartoll, X., Gil, J., & Ramos, R. (2019). Temporary employment, work stress and mental health before and after the Spanish economic recession. *International Archives of Occupational and Environmental Health*, 92, 1047–1059. https://doi.org/10.1007/s00420-019-01443-2
- Benz, M., & Frey, B. S. (2008). Being independent is a great thing: Subjective evaluations of self-employment and hierarchy. *Economica*, 75, 362–383. https://doi.org/10.1111/j.1468-0335.2007.00594.x
- Bérastégui, P. (2021). Exposure to psychosocial risk factors in the gig economy: A systematic review. (ETUI Research Paper, Report 2021.01). SSRN. https://doi.org/10.2139/ssrn.3770016
- Berger, T., Frey, C. B., Levin, G., & Danda, S. R. (2019). Uber happy? Work and wellbeing in the gigeconomy. *Economic Policy*, 34(99), 429–477. https://doi.org/10.1093/epolic/eiz007
- Binder, M., & Blankenberg, A. K. (2021). Self-employment and subjective well-being. In K. F. Zimmermann, K. F. (Ed.), Handbook of labor, human resources and population economics. Springer. https://doi.org/10.2139/ssrn.3768347

- Binder, M., & Coad, A. (2013). Life satisfaction and self-employment: A matching approach. Small Business Economics, 40(4), 1009–1033. https://doi.org/10.1007/s11187-011-9413-9
- Blanchard, O., & Landier, A. (2002). The perverse effects of partial labour market reform: Fixed-term contracts in France. *The Economic Journal*, 112(480), F214–F244. https://doi.org/10.1111/1468-0297.00047
- Blanchflower, D. G., & Oswald, A. (1998). What makes an entrepreneur? *Journal of Labor Economics*, 16(1), 26–60. https://doi.org/10.1086/209881
- Blanchflower, D. G. (2000). Self-employment in OECD countries. *Labour Economics*, 7, 471–505. https://doi.org/10.1016/S0927-5371(00)00011-7
- Böheim, R., & Mühlberger, U. (2009). Dependent forms of self-employment in the UK. Identifying workers on the border between employment and self-Employment. *Journal of Labour Market Research*, 42(2), 182–195. https://doi.org/10.1007/s12651-009-0014-x
- Cacioppo, J. T., & Patrick, B. (2008). Loneliness: Human nature and the need for social connection. W. W. Norton & Company.
- Carrieri, V., Di Novi, C., Jacobs, R., & Robone, S. (2012). Well-Being and psychological consequences of temporary contracts: The case of younger Italian employees (CHE Research Paper 79). Centre for Health Economics, University of York.
- Cheng, G.H.-L., & Chan, D.K.-S. (2008). Who suffers more from job insecurity? A meta-analytic review. *Applied Psychology: An International Review*, 57(2), 272–303. https://doi.org/10.1111/j.1464-0597.2007.00312.x
- Clark, A., Diener, E., Georgellis, Y., & Lucas, R. E. (2008). Lags and leads in life satisfaction: A Test of the Baseline Hypothesis. *The Economic Journal*, *118*(529), 222–243. https://doi.org/10.1111/j.1468-0297.2008.02150.x
- Cottini, E., & Lucifora, C. (2013). Mental health and working conditions in Europe. *ILR Review*, 66(4), 958–988. https://doi.org/10.1177/001979391306600409
- D'Amuri, F., & Marcucci, J. (2017). The predictive power of Google searches in forecasting US unemployment. *International Journal of Forecasting*, 33(4), 801–816. https://doi.org/10.1016/j.ijforecast.2017.03.004
- Danner, D. D., Friessen, W. V., & Snowdon, D. A. (2001). Positive emotions in early life and longevity: Findings from the Nun study. *Journal of Personality and Social Psychology*, 80(5), 804–813. https://doi.org/10.1037/0022-3514.80.5.804
- Dawson, C., Veliziotis, M., & Hopkins, B. (2017). Temporary employment, job satisfaction and subjective wellbeing. *Economic and Industrial Democracy*, 38(1), 69–98. https://doi.org/10.1177/0143831X14559781
- Dawson, C., Veliziotis, M., Pacheco, G., & Webber, D. J. (2015). Is temporary employment a cause or consequence of poor mental health? A panel data analysis. *Social Science & Medicine*, 134, 50–58. https://doi.org/10.1016/j.socscimed.2015.04.001
- De Cuyper, N., & De Witte, H. (2006). The impact of job insecurity and contract type on attitudes, wellbeing and behavioural reports: A psychological contract perspective. *Journal of Occupational and Organizational Psychology*, 79(3), 395–409. https://doi.org/10.1348/096317905X53660
- De Cuyper, N., & De Witte, H. (2007). Job insecurity in temporary versus permanent workers: Associations with attitudes, well-being, and behaviour. *Work & Stress, 21*(1), 65–84. https://doi.org/10.1080/02678370701229050
- De Stefano, V. (2016). The rise of the just-in-time workforce: On-demand work, crowdwork, and labor protection in the gig-economy. ILO Conditions of Work and Employment Series No. 71. International Labour Organization. https://doi.org/10.2139/ssrn.2682602
- De Witte, H. (1999). Job insecurity and psychological well-being: review of the literature and exploration of some unresolved issues. *European Journal of Work and Organizational Psychology*, 8(2), 155–177. https://doi.org/10.1080/135943299398302

- De Witte, H. (2005). Job insecurity: Review of the international literature on definitions, prevalence, antecedents and consequences. South African Journal of Industrial Psychology, 31(4), 1–6. https://doi.org/10.4102/sajip.v31i4.200
- De Witte, H., & Näswall, K. (2003). Objective versus subjective job insecurity: Consequences of temporary work for job satisfaction and organizational commitment in four European countries. *Economic and Industrial Democracy*, 24, 149–188. https://doi.org/10.1177/0143831X03024002002
- Di Tella, R., MacCulloch, R., & Oswald, A. (2003). The macroeconomics of happiness. *Review of Economics and Statistics*, 85, 809–827. https://doi.org/10.1162/003465303772815745
- Drahokoupil, J., & Fabo, B. (2018). Outsourcing, offshoring and the deconstruction of employment: New and old challenges. In A. Serrano-Pascual & M. Jepsen (Eds.), *The deconstruction of employment as a political question: 'Employment' as a floating signifier* (pp. 33–61). Springer Verlag. https://doi.org/10.1007/978-3-319-93617-8_2
- Eichhorst W., Marx, P., & Wehner, C. (2017). Labor market reforms in Europe: Towards more flexicure labor markets? Journal for Labour Market Research, 51(3), 1–17. https://doi.org/10.1186/s12651-017-0231-7
- Eichhorst, W., & Tobsch, V. (2017). Risk of precariousness: Results from European Working Conditions Survey 2010 and 2015. Brussels. Economy and Scientific Policy European Parliament.
- Ertel, M., Pech, E., Ullsperger, P., Von Dem Knesebeck, O., & Siegrist, J. (2005). Adverse psychosocial working conditions and subjective health in freelance media workers. *Work Stress*, 19(3), 293–299. http://dx.doi.org/10.1080/02678370500307289
- Escudero-Castillo, I., Mato Díaz, F.J. & Rodriguez-Alvarez, A. (2023). Effects of precarious work on mental health: Evidence from Spain. *Applied Economics*, 55(14), 1603–1620. https://doi.org/10.1080/00036846.2022.2098241
- Eurofound. (2017). *Exploring self-employment in the European Union*. Publications Office of the European Union, Luxembourg. https://doi.org/10.2806/651917
- European Commission. (2017). Second phase consultation of Social Partners under Article 154 TFEU on a possible revision of the Written Statement Directive (Directive 91/533/EEC) in the framework of the European Pillar of Social Rights {C(2017) 6121 final}. European Commission, Brussels.
- Fabrin-Petersen, R. (2022). The effects of non-standard employment on subjective well-being: A meta-analytic review. Advances in Methodology and Statistics, 19(1), 1–39. https://doi.org/10.51936/uspx6788
- Felgueroso, F., García-Pérez, J. I., Jansen, M., & Troncoso-Ponce, D. (2017). *Recent trends in the use of temporary contracts in Spain*. Studies on the Spanish Economy eee2017-25, FEDEA.
- Ferrer-i-Carbonell, A., & Frijters, P. (2004). How important is methodology for the estimates of the determinants of happiness? The Economic Journal, 114, 641–659. https://doi.org/10.1111/j.1468-0297.2004.00235.x
- Gevaert, J., De Moortel, D., Wilkens, M., & Vanroelen, C. (2018). What's up with the self-employed? A cross-national perspective on the self-employed's work-related mental well-being. SSM – Population Health, 4, 317–326. https://doi.org/10.1016/j.ssmph.2018.04.001
- Glavin, P., & Schieman, S. (2022). Dependency and hardship in the gig economy: The mental health consequences of platform work. Socius, 8. https://doi.org/10.1177/23780231221082414
- GOVUP & Adigital. (2017). Trabajo en plataformas digitales: Análisis y propuestas de regulación. Adigital, Madrid.
- Greenhalgh, L., & Rosenblatt, Z. (2010). Evolution of research on job insecurity. International Studies of Management & Organization, 40(1), 6–19. https://doi.org/10.2753/IMO0020-8825400101
- Gross, S.-A., Musgrave, G., & Janciute, L. (2018). Well-being and mental health in the gig economy: Policy perspectives on precarity (Vol. 4). University of Westminster Press. https://doi.org/10.16997/book32
- Helbling, L., & Kanji, S. (2018). Job insecurity: Differential effects of subjective and objective measures on life satisfaction trajectories of workers aged 27–30 in Germany. *Social Indicators Research*, 137, 1145–1162. https://doi.org/10.1007/s11205-017-1635-z

- Hyytinen, A., & Ruuskanen, O. P. (2007). Time use of the self-employed. *Kyklos*, *60*(1), 105–122. https://doi.org/10.1111/j.1467-6435.2007.00361.x
- International Labour Office. (2015, February 16–19). Non-standard forms of employment. Report for discussion at the Meeting of Experts on Non-Standard Forms of Employment, Geneva.
- International Labour Office. (2016). Non-standard employment around the world: Understanding challenges, shaping prospects. Geneva.
- International Labour Office. (2018). Digital labour platforms and the future of work: Towards decent work in the online world. Geneva.
- Inanc, H. (2018). Unemployment, temporary work, and subjective well-being: The gendered effect of spousal labor market insecurity. *American Sociological Review*, 83(3), 536–566. https://doi.org/10.1177/0003122418772061
- Kara, A., & Petrescu, M. (2018). Self-employment and its relationship to subjective well-being. International Review of Entrepreneurship, 16(1), 115–140.
- Karabchuk, T., & Soboleva, N. (2020). Temporary employment, informal work and subjective well-being across Europe: Does labor legislation matter? *Journal of Happiness Studies*, 21, 1879–1901. https://doi.org/10.1007/s10902-019-00152-4
- Kässi, O., & Lehdonvirta, V. (2016). *Building the online labour index: a tool for policy and research*. CSCW and the Sharing Economy: The Future of Platforms as Sites of Work Collaboration and Trust. Association for Computing Machinery.
- Kässi, O., & Lehdonvirta, V. (2018). Online labour index: Measuring the online gig economy for policy and research. *Technological Forecasting and Social Change*, 137, 241–248. https://doi.org/10.1016/j.techfore.2018.07.056
- Kautonen, T., Down, S., Welter, F., Vainio, P., Palmroos, J., Althoff, K., & Kolb, S. (2010). 'Involuntary self-employment' as a public policy issue: A cross-country European review. *International Journal of Entrepreneurial Behaviour and Research*, 16(2), 112–129. https://doi.org/10.1108/13552551011027002
- Kawaguchi, D. (2008). Self-employment rents: Evidence from job satisfaction scores. *Hitotsubashi Journal of Economics*, 49(1), 35–45.
- Keim, A. C., Landis, R. S., Pierce, C. A., & Earnest, D. R. (2014). Why do employees worry about their jobs? A meta-analytic review of predictors of job insecurity. *Journal of Occupational Health Psychology*, 19(3), 269–290. https://doi.org/10.1037/a0036743
- Keith, M. G., Harms, P. D., & Long, A. C. (2020). Worker health and well-being in the gig economy: A proposed framework and research agenda. In P. L. Perrewé, P. D. Harms, & C.-H. Chang (Eds.), *Entrepreneurial and small business stressors, experienced stress, and well being*. Emerald Publishing. https://doi.org/10.1108/S1479-35552020000018002
- Klein Hesselink, D. J., & van Vuuren, T. (1999). Job flexibility and job insecurity: The Dutch case. European Journal of Work and Organizational Psychology, 8, 273–293. https://doi.org/10.1080/135943299398366
- Millan, J. M, Hessels, J., Thurik, R., & Aguado, R. (2013). Determinants of job satisfaction: A European comparison of self-employed and paid employees. *Small Business Economics*, 40, 651–670. https://doi.org/10.1007/s11187-011-9380-1
- Monereo, J. (2016). Transformaciones del trabajo y futuro del derecho de trabajo en la "era de la flexibilidad". Derecho de las relaciones laborales, 2016(5), 444–493.
- Nordenmark, M., Vinberg, S., & Strandh, M. (2012). Job control and demands, work-life balance and wellbeing among self- employed men and women in Europe. *Vulnerable Groups Inclusion*, 3(1), Article 18896. https://doi.org/10.3402/vgi.v3i0.18896
- OECD. (2015). Non-standard work, job polarisation and inequality, in In It Together: why less inequality benefits all. OECD Publishing, Paris. https://doi.org/10.1787/9789264235120-7-en
- OECD. (2019). OECD employment outlook 2019: The future of work. OECD Publishing, Paris. https://doi.org/10.1787/9ee00155-en

- OECD/ILO/European Union. (2023). Handbook on measuring digital platform employment and work. OECD Publishing, Paris. https://doi.org/10.1787/0ddcac3b-en
- Pesole, A., Urzi Brancati, M. C., Fernandez-Macias, E., Biagi, F., & Gonzalez Vazquez, I. (2018). Platform workers in Europe evidence from the COLLEEM survey (EUR 29275 EN). Publications Office of the European Union, Luxembourg. https://doi.org/10.2760/742789
- Ponzo, M. (2011). Status occupazionale e benessere soggettivo in Italia. QA Rivista dell Associazione Rossi-Doria, 3(3), 75–106. https://doi.org/10.3280/QU2011-003004
- Probst, T. M. (2008). Job insecurity. In C. L. Cooper & J. Barling (Eds.), The SAGE handbook of organizational behavior: Volume I-micro approaches. SAGE Publications Ltd. https://doi.org/10.4135/9781849200448
- Quesnel-Vallee, A., DeHaney, S., & Ciampi, A. (2010). Temporary work and depressive symptoms: A prospective score analysis. Social Science and Medicine, 70, 1982–1987. https://doi.org/10.1016/j.socscimed.2010.02.008
- Robone, S., Jones, A. M., & Rice, N. (2011). Contractual conditions, working conditions, health and wellbeing in the British Household Panel Survey. *The European Journal of Health Economics*, 12, 429–444. https://doi.org/10.1007/s10198-010-0256-0
- Rodríguez, E. (2002). Marginal employment and health in Britain and Germany: Does unstable employment predict health? *Social Science and Medicine*, 55(6), 963–979. https://doi.org/10.1016/S0277-9536(01)00234-9
- Russo, C., & Terraneo, M. (2020). Mental well-being among workers: A crossnational analysis of job insecurity impact on the workforce. *Social Indicators Research*, *152*, 421–442. https://doi.org/10.1007/s11205-020-02441-5
- Scherer, S. (2009). The social consequences of insecure Jobs. Social Indicators Research, 93, 527–547. https://doi.org/10.1007/s11205-008-9431-4
- Schulze Buschoff, K., & Schmidt, C. (2009). Adapting labour law and social security to the needs of the 'new self-employed' – comparing the UK, Germany and the Netherlands. *Journal of European Social Policy*, 19(2), 147–159. https://doi.org/10.1177/0958928708101867
- Schumann, P., & Kuchinke, L. (2020). Do(n't) worry, it's temporary: The effects of fixed-term employment on affective well-being. *Journal of Happiness Studies*, 21, 2557–2582. https://doi.org/10.1007/s10902-019-00194-8
- Schwellnus, C., Geva, A., Pak, M., & Veiel, R. (2019). Gig economy platforms: Boon or Bane? (OECD Economics Department Working Papers, No. 1550). OECD Publishing, Paris. https://doi.org/10.1787/fdb0570b-en
- Silla, I., Gracia, F. J., & Peiro, J. M. (2005). Job insecurity and health-related outcomes among different types of temporary workers. *Economic and Industrial Democracy*, 26, 89–117. https://doi.org/10.1177/0143831X05049404
- Stone, K. V. (2006). Legal protections for atypical employees: employment law for workers without workplaces and employees without employers. *Berkeley Journal of Employment and Labour Law*, 27(2), 251–286. https://doi.org/10.15779/Z38Q342
- Sverke, M., Gallagher, D. G., & Hellgren, J. (2000). Alternative work arrangments: Job stress, well-being, and work attitudes among employees with different employment contracts. In K. Isaksson, L. Hogstedt, C. Eriksson & T. Theorell (Eds.), *Health effects of the new labour market*. Plenum.
- Sverke, M., Hellgren, J., & Naswall, K. (2002). No security: A meta-analysis and review of job insecurity and its consequences. *Journal of Occupational Health Psychology*, 7(3), 242–264. https://doi.org/10.1037/1076-8998.7.3.242
- Syrett, M. (2016). Independent professionals (IPros) and well-being: An emerging focus for research? Journal of Management and Organization, 22(6), 817–825. https://doi.org/10.1017/jmo.2016.54
- Valletta, R. G., Bengali, L., & van der List, C. (2020). Cyclical and market determinants of involuntary parttime employment. *Journal of Labor Economics*, 38(1), 67–93. https://doi.org/10.1086/704496

- Virtanen, J., Vahtera, M., Kivimäki, J., Pentii, J., & Ferrie, J. (2002). Employment security and health. *Journal of Epidemiological Community Health*, 56, 569–574. https://doi.org/10.1136/jech.56.8.569
- Vučeković, M., Avlijaš, G., Marković, M. R., Radulović, D., Dragojević, A., & Marković, D. (2023). The relationship between working in the "gig" economy and perceived subjective well-being in Western Balkan countries. *Frontiers in Psychology*, 14. https://doi.org/10.3389/fpsyg.2023.1180532
- Wang, S., Li, L. Z., & Coutts, A. (2022). National survey of mental health and life satisfaction of gig workers: The role of loneliness and financial precarity. *BMJ Open*, *12*, Article e066389. https://doi.org/10.1136/bmjopen-2022-066389
- Westerveld, M. (2012). The "new" self-employed: An issue for social policy? European Journal of Social Security, 14(3), 156–173. https://doi.org/10.1177/138826271201400301
- Wood, A. J., Graham, M., Lehdonvirta, V., & Hjorth, I. (2018). Good gig, bad gig: Autonomy and algorithmic control in the global gig economy. Work, Employment and Society, 33(1), 56–75. https://doi.org/10.1177/0950017018785616
- Wu, Q. M. (2023). Employment precarity, COVID-19 risk, and workers' well-being during the pandemic in Europe. Work and Occupations, 50(2), 188–211. https://doi.org/10.1177/07308884221126415

APPENDIX

Appendix A. Main definitions and data characteristics

Table A1. Description of dependent variables: Self-defined current (main) economic status

Main job	Respondents declare their main current activity status on the basis of how most time is spent. Persons who work simultaneously in their own professional practice and for a public or private employer should be classified according to the status where they work the longer number of hours. If the person combines different part-time jobs as an employee that result in the equivalent hours of a full-time job, the person should consider herself as an employee working full-time. The same holds if the person has a main job as an employee and an additional "small" (in terms of hours) second job as a self- employed person, both jobs being remunerated in what would be considered, together, as the remuneration for a full-time equivalent role.
Self-employed	In the ECV the self-employed are defined as persons who work in their own business, professional practice or farm for the purpose of earning a profit. Among the self-employed, those who employ paid employees are defined as "employers", while those without paid employees are defined as "own-account workers". Self-employment is becoming more prevalent in the EU, and it is becoming more heterogeneous in its nature. The types of occupation in which the self-employed are engaged have diversified. Many new self-employed are involved in the service sector, but the skill level of these jobs varies widely from low-paid jobs on temporary contracts to high-paid and specialised jobs in banking and financial services and information technology. The traditional self- employed person running a business, perhaps with a few employees, is joined by people in casual work or involved with sub-contracting.
Self-employed without employees (Own-account workers)	Own-account workers are defined as persons who work in their own business, professional practice or farm for the purpose of earning a profit and who do not employ any other person. A freelancer should be classified as self-employed, although a person who has been regularly retained by a single employer for some time may also be regarded as an employee.
Temporary employees	Temporary employees are defined as workers who work for a public or private employer receiving compensation in the form of wages, salaries, fees, etc. but both the employer and the employee understand that the termination of the job is determined by objective conditions such as reaching a certain date, completion of an assignment or return of another employee who has been temporarily replaced. The condition for its termination is generally mentioned in the contract.

	Employe	ed individ	uals (N =	11,377)	Tota	al sample	(N = 16,	527)
	Mean	SD	Min	Max	Mean	SD	Min	Max
Permanent workers	0.665	0.472	0	1	0.458	0.498	0	1
Temporary workers	0.222	0.416	0	1	0.153	0.360	0	1
Self-employed (own-account workers)	0.112	0.316	0	1	0.077	0.267	0	1
Unemployed	-	-	0	0	0.155	0.362	0	1
Inactive	-	-	0	0	0.157	0.364	0	1
Number of children	0.632	0.861	0	5	0.563	0.840	0	5
Ln (Household size)	1.385	0.306	0.693	2.485	1.374	0.310	0.693	2.639
Ln (Household income)	9.718	0.650	0.693	11.893	9.564	0.926	0.336	12.007
Household with difficulties to make ends meet	0.483	0.500	0	1	0.542	0.498	0	1
Female	0.472	0.499	0	1	0.504	0.500	0	1
Married	0.596	0.491	0	1	0.579	0.494	0	1
Divorced	0.076	0.264	0	1	0.078	0.269	0	1
Widowed	0.011	0.105	0	1	0.018	0.132	0	1
16–24 years	0.038	0.192	0	1	0.056	0.230	0	1
25–34	0.158	0.364	0	1	0.147	0.354	0	1
35–44	0.286	0.452	0	1	0.244	0.430	0	1
45–54	0.308	0.462	0	1	0.283	0.450	0	1
55–64	0.210	0.407	0	1	0.270	0.444	0	1
Primary education	0.063	0.243	0	1	0.102	0.303	0	1
Lower secondary	0.248	0.432	0	1	0.279	0.448	0	1
Upper secondary	0.243	0.429	0	1	0.244	0.430	0	1
Tertiary	0.447	0.497	0	1	0.375	0.484	0	1
Ln (Unemployment rate)	2.603	0.263	2.300	3.163	2.628	0.276	2.300	3.163
Ln (Regional income)	10.268	0.144	9.953	10.465	10.256	0.147	9.953	10.465
Blue collar occupation	0.535	0.499	0	1	0.597	0.490	0	1

Table A2. Sample characteristics (source: own elaboration (ECV-2018))

Table A3. Well-being variables

Variable	Question	Original scale	Transformation
Self-perceived general health	How is your health in general? (The respondent should take into account the different dimensions of health, i.e. physical, social and emotional functioning and biomedical signs and symptoms).	1 Very good 2 Good 3 Fair 4 Bad 5 Very bad	5 Very good 4 Good 3 Fair 2 Bad 1 Very bad
Overall life satisfaction	Overall, how satisfied are you with your current life? (The respondent should take into account that the term "life" is intended as all areas of a person's life at the time of the survey).	From 0 (Not at all satisfied) to 10 (Completely satisfied)	From 1 (Not at all satisfied) to 11 (Completely satisfied)
Satisfaction with financial situation	Overall, how satisfied are you with the financial situation of your household? (The respondent should take into account income adequacy, level of savings, capacity to pay back debt and money owed, the ability to meet large emergency expenses, level of assets for the entire household).	From 0 (Not at all satisfied) to 10 (Completely satisfied)	From 1 (Not at all satisfied) to 11 (Completely satisfied)
Satisfaction with personal relationships	Overall, how satisfied are you with people you often spend time with? (The respondent should take into consideration his/her relationships with family, friends, colleagues, neighbors, etc.).	From 0 (Not at all satisfied) to 10 (Completely satisfied)	From 1 (Not at all satisfied) to 11 (Completely satisfied)
Satisfaction with time use (amount of leisure time)	Overall, how satisfied are you with the amount of time you have to do things that you like? (The respondent should take into consideration time for hobbies, leisure, time off work)	From 0 (Not at all satisfied) to 10 (Completely satisfied)	From 1 (Not at all satisfied) to 11 (Completely satisfied)
Satisfaction with job	Overall, how satisfied are you with your current job? (The respondent should take into consideration tasks performed, workplace atmosphere, pay, hours, etc.)	From 0 (Not at all satisfied) to 10 (Completely satisfied)	From 1 (Not at all satisfied) to 11 (Completely satisfied)
Being happy	During the last four weeks how often have you been happy?	1 All of the time 2 Most of the time 3 Some of the time 4 A little of the time 5 None of the time	5 All of the time 4 Most of the time 3 Some of the time 2 A little of the time 1 None of the time

Note: For the estimation of Equation (1), the different well-being measures are considered to encompass a cardinality-type classification. The assumption of cardinality has the advantage of producing coefficients that can be directly interpreted as marginal effects (Ferrer-i-Carbonell & Frijters, 2004).



Figure A1. Self-employment by personal characteristics (% over total employment): 2000–2018 (source: own elaboration (European Labour Force Survey, Eurostat, Population aged 15–64 years))



Figure A2. Temporary employment by personal characteristics (% over total employment): 2000–2018 (source: own elaboration (European Labour Force Survey, Eurostat, Population aged 15–64 years))

Appendix B. Robustness checks

To check our results robustness, we have performed some extra analysis. First, we distinguish between full-time and part-time workers. In the ECV, full-time workers are defined as those working more than 30 weekly hours in their main job. In our sample, 5.8% (74 individuals) and 25.4% (642 individuals) of self-employed and temporary employees, respectively, are part-time workers. As the sample of part-time self-employed workers is too small (only 74 individuals), we have replicated the estimates considering the difference between "gig-related" full-time vs. "gig-related" part-time in the first set of estimations and between "self", "full-time temporary employee", "part-time temporary employee" in the second.

The results are shown in Table B1 and B2 for employed people and the entire sample, respectively. The general results remain: the negative effect of the work arrangements most common in the gig economy on individuals' well-being only comes from self-employment. Note that when we consider the broader category "gig-related employment" the main effect

is mainly driven by full-time workers. We cannot be sure if this is because of the intrinsic characteristics of this type of (full-time) workers or due to a composition effect, as the negative effect is driven by self-employment and in our sample the part-time self-employed workers are under-represented (5.8%).

Finally, we have replicated the above estimates for two additional restricted subsamples. As having an internet connection is essential to participate in the digital platform economy, we have considered a first restricted subsample of workers excluding those self-employed and temporary employees who declare not having an internet connection for personal use at home (the internet access can be via Smartphone, other wireless handheld device such as tablet or others, video games console, laptop, desktop computer, TV, etc.). Among the internet activities for personal use, the ECV includes buying/selling goods or services, as well as others (social networking, sending/receiving emails, using services related to travel and accommodation, creating web pages, blogs, Internet banking, reading or downloading online music, video, news, looking for information, telephoning or making video calls, taking part in online consultations or voting on civic or political issues). This restricted subsample corresponds to 95.4% and 93.9% of the original sample of self-employed and temporary employees, respectively (Tables B1 and B2 "Restricted definition 1: Internet").

As the type of activities related to the digital platform economy is mainly developed in the service sector, we have considered a second set of estimates with only self and temporary workers employed in the service sector (and having an internet connection at home). This restricted definition corresponds to 64.3% and 67.3% of the original sample of self-employed and temporary employees, respectively (a dummy for the remaining self-employed and temporary workers is included among control variables to keep permanent employees as the reference category). Results are shown in Tables B1 and B2 ("Restricted definition 2: Service sector + Internet"). As can be seen, the main results hold, although the significance level is lower when the definition of gig economy related work is restricted to the service sector.

		S	elf-report	ed healt	h				Happin	ess		
	Orig sam		Restri definit Inter	ion 1:	Restricted definition 2: Service sector + Internet		Original sample		Restricted definition 1: Internet		Restricted definition 2: Service sector + Internet	
	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)
GIG_Emp_	-2.514***	-	-2.492***	-	-5.409+	-	-2.088***	-	-2.273***	-	-5.287*	-
full	[0.644]		[0.754]		[2.833]		[0.483]		[0.569]		[2.178]	
GIG_Emp_	2.819	-	3.334	-	7.306	-	0.083	-	-0.311	-	2.139	-
part	[2.147]		[2.580]		[5.156]		[1.709]		[2.059]		[4.103]	
Self_		-4.357**	-	-4.208*	-	-8.734+	-	-2.584**	—	-2.710*	-	-5.661*
employed		[1.486]		[1.634]		[4.990]		[0.995]		[1.102]		[2.792]
Temporary_	-	-1.065	-	-1.143	-	-1.584	-	-0.229	-	-0.138	-	0.323
Emp_full		[0.766]		[0.779]		[2.205]		[0.534]		[0.542]		[1.281]

 Table B1. Effects of the digital platform work related on individual well-being by workhour

 (Employed persons)

IV-1)riginal ample	Post									
	umpie	defini	icted tion 1: rnet	Restricted definition 2: Service sector + Internet		Original sample		Restricted definition 1: Internet		Restricted definition 2: Service sector + Internet	
	(a) IV-2 (b) IV-1 ^(a)	IV-2 (b)	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 (b)	IV-1 ^(a)	IV-2 ^(b)
Temporary	2.884	-	3.505	-	8.453	-	1.278	-	1.267	-	2.387
Emp_part	[3.016]	[3.594]		[6.866]		[2.120]		[2.552]		[4.060]
	Life	satisfaction					Fin	isfaction	·		
GIG_Emp2.91	i*** –	-3.004***	-5.316**	-	-5.316**	-1.879***	-	–1.757***	-	-3.629*	-
full [0.6	4]	[0.799]	[2.017]		[2.017]	[0.438]		[0.499]		[1.746]	
GIG_Emp5.1	2* –	-5.270+	-2.41	-	-2.41	0.403	-	0.894	-	3.252	-
part [2.3	4]	[2.768]	[3.746]		[3.746]	[1.495]		[1.732]		[3.214]	
Self	-8.831		-	-15.920*	-	-	-5.364***	-	-4.868**	-	-9.146*
employed	[2.887	1		[6.872]			[1.611]		[1.618]		[4.196]
Temporary	1.335	-	-	2.586	-	-	1.032	-	0.962	-	2.023
Emp_full	[1.459	1		[3.121]			[0.838]		[0.773]		[1.927]
Temporary	-8.04) _	-	-4.574	-	-	-0.509	-	0.213	-	2.789
Emp_part	[5.899	1		[9.768]			[3.396]		[3.664]		[5.968]
Per	Personal relationships satisfaction						Le	isure satis	sfaction		
GIG Emp -1.35	7** –	-1.397**	-2.386+	-	-2.386+	-1.492**	-	-1.328*	-	-2.954	_
full [0.4	1]	[0.525]	[1.218]		[1.218]	[0.520]		[0.592]		[1.913]	
GIG_Emp1.8	91 –	-2.299	-0.482	-	-0.482	2.436	-	2.76	-	5.414	_
part [1.4	9]	[1.793]	[2.194]		[2.194]	[1.719]		[1.998]		[3.442]	
Self -	-5.654	* –	-	-9.044+	-	-	-2.259*	_	-1.954	-	-4.615
employed	[2.042	1		[4.717]			[1.143]		[1.222]		[3.463]
Temporary	2.320	-	-	4.206*	-	-	-1.276*	-	-1.390*	-	-2.099
Emp_full	[1.040	1		[2.126]			[0.602]		[0.601]		[1.554]
Temporary	-3.44	5 –	-	-2.120	-	-	2.436	-	2.671	-	6.530
Emp_part	[4.189	1		[6.536]			[2.224]		[2.591]		[4.704]
		Job sati	sfaction	1	<u> </u>		1	1			
GIG_Emp1.39	7** –	-1.226*	-	-2.230+							
full [0.43	6]	[0.481]		[1.247]							
GIG_Emp1.3	5 –	-0.999	-	0.737							
part [1.50	7]	[1.689]		[2.325]							
Self –	-4.634	* _	-4.310**	-	-7.940*						
employed	[1.587		[1.628]		[3.556]						
Temporary	0.376	-	0.236	-	0.600						
Emp_full	[0.828	1	[0.781]		[1.684]						
Temporary	-3.404		-3.254	-	-0.300						
Emp_part	[3.286		[3.626]		[5.065]						
N 11,3	7 11,37	11,163	11,163	11,377	11,377	11,377	11,377	11,163	11,163	11,377	11,377

Standard errors in brackets. + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001. Reference category: Permanent workers.

^(a) Self-employment and temporary employment are jointly considered in the variable *Gig*; ^(b) Self-employment and temporary employment appear separately. Self-employed refers to own-account workers.

Restricted definition 1: Self-employed and temporary employees are defined as self-employed (own-account workers) and temporary employees who have an internet connection at home for several internet activities, including buying/selling goods and services. *Restricted definition 2*: Self-employed and temporary employees are defined as self-employed (own-account workers) and temporary employees in the service sector who have an internet connection.

Controls: gender, age, educational level, marital status, number of children, household size, household income, household with difficulties to make ends meet, white/blue collar, regional unemployment rate, regional mean income. A dummy of the remaining self and temporary workers is included among control variables for estimates based on the second restricted definition. *Instruments*: Google trends data on Just Eat, Uber, and Airbnb for Model IV-1. Just Eat, Uber, Airbnb, and population size for IV-2.

 Table B2. Effects of the work-related with the digital platform economy on individual well-being by workhour (Total sample)

	Self-reported health							Happiness							
	Original sample		Restricted definition 1: Internet		Restricted definition 2: Service sector + Internet		Original sample		Restricted definition 1: Internet		Restricted definition 2: Service sector + Internet				
	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)			
GIG_Emp_full	-3.416***	-	-3.577***	-	-7.848+	-	-3.603***	-	-4.007***	-	-9.558*	-			
	[0.837]	-	[1.047]	-	[4.223]	-	[0.849]	-	[1.092]	-	[4.678]	-			
GIG_Emp_ part	1.383	-	1.657	-	6.032	-	0.764	-	0.313	-	4.037	-			
	[2.611]	-	[3.201]	-	[6.079]	-	[2.718]	-	[3.415]	-	[6.848]	-			
Self_ employed	-	-6.341**	-	-6.593**	-	-13.211+	-	-4.629**	-	-4.799*	-	-9.385+			
	-	[2.028]	-	[2.411]	-	[7.170]	-	[1.714]	-	[2.049]	-	[5.481]			
Temporary_ Emp_full	-	-0.971	-	-1.089	-	-1.022	-	-0.577	-	-0.491	-	0.228			
	-	[1.016]	-	[1.047]	-	[2.809]	-	[0.869]	-	[0.900]	-	[2.152]			
Temporary_	-	0.907	-	1.104	-	5.942	-	2.611	-	3.081	-	5.549			
Emp_part	-	[3.968]	-	[4.930]	-	[8.342]	-	[3.384]	-	[4.239]	-	[6.482]			
Unemployed	-1.166**	-1.088*	-1.156*	-1.095+	-1.402	-1.033	-1.338**	-0.699	-1.464**	-0.656	-2.020	-0.605			
	[0.422]	[0.514]	[0.525]	[0.643]	[1.131]	[1.120]	[0.433]	[0.434]	[0.554]	[0.548]	[1.259]	[0.861]			
Inactive	-1.279***	-1.282**	-1.269**	-1.284*	-1.481	-1.213	-1.073**	-0.552	-1.189*	-0.513	-1.681	-0.458			
	[0.385]	[0.491]	[0.479]	[0.612]	[1.027]	[1.057]	[0.396]	[0.414]	[0.506]	[0.521]	[1.144]	[0.812]			
Life satisfaction								Financial satisfaction							
	-5.189***	-	-5.677**	-	-10.277*	-	-2.774***	-	-2.708***	-	-5.664+	-			
GIG_Emp_full	[1.371]	-	[1.760]	-	[4.652]	-	[0.660]	-	[0.801]	-	[2.996]	-			
GIG_Emp_ part	-9.707*	-	-10.142+	-	-5.040	-	-0.336	-	0.280	-	3.427	-			
	[4.202]	-	[5.295]	-	[6.753]	-	[2.080]	-	[2.459]	-	[4.348]	-			
Self_ employed	-	-15.940**	-	-17.202*	-	-31.294+	-	-7.668**	-	-7.414**	-	-14.031+			
	-	[5.663]	-	[6.894]	-	[16.023]	-	[2.428]	-	[2.688]	-	[7.198]			
Temporary_ Emp_full	-	2.515	-	2.098	-	4.874	-	1.171	-	1.051	-	2.489			
	-	[2.792]	-	[2.976]	-	[6.312]	-	[1.218]	-	[1.166]	-	[2.866]			
Temporary_	-	-15.300	-	-17.048	-	-11.712	-	-1.963	-	-1.382	-	1.651			
Emp_part	-	[10.871]	-	[13.916]	-	[18.776]	-	[4.820]	-	[5.550]	-	[8.503]			
Unemployed	-3.018***	-3.145*	-3.098***	-3.426+	-3.170*	-3.523	-1.407***	-1.330*	-1.303**	-1.259+	-1.445+	-1.265			
	[0.687]	[1.409]	[0.879]	[1.815]	[1.250]	[2.510]	[0.334]	[0.619]	[0.402]	[0.721]	[0.804]	[1.136]			
Inactive	-2.491***	-2.868*	-2.563**	-3.107+	-2.622*	-3.135	-0.905**	-0.960	-0.810*	-0.879	-0.931	-0.858			
	[0.626]	[1.345]	[0.801]	[1.726]	[1.136]	[2.368]	[0.305]	[0.592]	[0.367]	[0.686]	[0.730]	[1.072]			

	Self-reported health						Happiness						
	Original sample		Restricted definition 1: Internet		Restricted definition 2: Service sector + Internet		Original sample		Restricted definition 1: Internet		Restricted definition 2: Service sector + Internet		
	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	IV-1 ^(a)	IV-2 ^(b)	
Personal relationships satisfaction							Leisure satisfaction						
GIG_Emp_full	-2.356**	-	-2.584*	-	-4.361+	-	-1.941**	-	-1.753*	-	-3.674	-	
	[0.779]	-	[1.007]	-	[2.344]	-	[0.661]	-	[0.802]	-	[2.638]	-	
GIG_Emp_ part	-4.529+	-	-5.043+	-	-2.214	-	1.733	-	2.240	-	5.201	-	
	[2.387]	-	[3.009]	-	[3.348]	-	[2.046]	-	[2.415]	-	[3.750]	-	
Self_	-	-9.727**	-	-10.065*	-	-16.678+	-	-4.449**	-	-4.335*	-	-8.981+	
employed	-	[3.757]	-	[4.394]	-	[9.875]	-	[1.571]	-	[1.784]	-	[5.069]	
Temporary_	-	3.819*	-	3.622+	-	6.880+	-	-0.786	-	-0.968	-	-1.140	
Emp_full	-	[1.851]	-	[1.895]	-	[3.860]	-	[0.803]	-	[0.792]	-	[2.028]	
Temporary_	-	-7.558	-	-8.446	-	-6.356	-	0.544	-	0.519	-	4.622	
Emp_part	-	[7.219]	-	[8.840]	-	[11.347]	-	[3.051]	-	[3.607]	-	[5.933]	
Unemployed	-1.248**	-1.084	-1.310**	-1.205	-1.234+	-1.123	-0.074	-0.269	0.042	-0.274	0.023	-0.209	
	[0.394]	[0.934]	[0.504]	[1.152]	[0.633]	[1.515]	[0.333]	[0.394]	[0.400]	[0.470]	[0.709]	[0.795]	
Inactive	-1.099**	-1.143	-1.157*	-1.237	-1.084+	-1.111	0.050	-0.177	0.152	-0.177	0.146	-0.104	
	[0.359]	[0.892]	[0.459]	[1.096]	[0.576]	[1.429]	[0.304]	[0.376]	[0.365]	[0.448]	[0.644]	[0.750]	
N	16,527	16,527	16,313	16,313	16,527	16,527	16,527	16,527	16,313	16,313	16,527	16,527	

Standard errors in brackets. + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001. Reference category: Permanent workers.

^(a) Self-employment and temporary employment are jointly considered in the variable *Gig*; ^(b) Self-employment and temporary employment appear separately. Self-employed refers to own-account workers.

Restricted definition 1: Self-employed and temporary employees are defined as self-employed (own-account workers) and temporary employees who have an internet connection at home for several internet activities, including buying/selling goods and services. *Restricted definition 2*: Self-employed and temporary employees are defined as self-employed (own-account workers) and temporary employees in the service sector who have an internet connection.

Controls: gender, age, educational level, marital status, number of children, household size, household income, household with difficulties to make ends meet, white/blue collar, regional unemployment rate, regional mean income. A dummy of the remaining self and temporary workers is included among control variables for estimates based on the second restricted definition. *Instruments*: Google trends data on Just Eat, Uber, and Airbnb for Model IV-1. Just Eat, Uber, Airbnb, and population size for IV-2.